Author's response to reviews

Title: Peripheral blood monocyte-to-lymphocyte ratio at study enrolment predicts efficacy of the RTS,S malaria vaccine: analysis of pooled phase 2 clinical trial data

Authors:

George M Warimwe (george.warimwe@ndm.ox.ac.uk)
Helen A Fletcher (helen.fletcher@ndm.ox.ac.uk)
Ally Olotu (aolotu@kemri-wellcome.org)
Selidji T Agnandji (agnandjis@lambarene.org)
Adrian VS Hill (adrian.hill@ndm.ox.ac.uk)
Kevin Marsh (kmash@kemri-wellcome.org)
Philip Bejon (pbejon@well.ox.ac.uk)

Version: 2 Date: 30 May 2013

Author's response to reviews:

Title modified to: Peripheral blood monocyte-to-lymphocyte ratio at study enrolment predicts efficacy of the RTS,S malaria vaccine: analysis of pooled phase 2 clinical trial data

Revisions and additions made to comply with STROBE guidelines (STROBE checklist uploaded as additional material.)